Walton Smith Cruise Narrative

9/6/2010

0630—Depart the dock in Berwick and begin transit to B246 (-91.5 27.5)

Finish preparing the MOCNESS during transit (attach nets and cod ends)

9/7/2010

10:40 CDT arrive at station B246.  The weather conditions prevent MOCNESS deployment.  We mounted the DVS acoustic instrument (2400 kHz) on the CTD frame for a cast.  

11:30 CDT we conducted a CTD cast and saw no distinct signal with the Aquatracka fluorometer at depth, but a great deal of noise in the signal in the upper 75 m of the water column.  We fired 3 bottles, one just below the peak in chl fuorescence, one in the peak, and one above the peak in the mixed layer at approximately 10 m.  We take nutrient, plankton, and size fractionated chl samples from these depths and run some of the water through the FlowCAM to get an immediate look at the small plankton.  There were very few plankton in the samples.  The altimeter on the CTD was very unstable and we removed the DVS in case it was interfering with the signal.

Because the weather still prevented MOCNESS deployment, we moved to station S8,T22 to do a CTD cast and to continue down the T22 line doing CTD casts.

15:18 CDT On station S8,T22.  We conducted a CTD cast.  The altimeter was still unstable and we wonder if the Aquatracka is affecting it.  There was a very tiny hint of a signal on the Aquatracka and a corresponding dip in oxygen at 1100 and 1150, but these were extremely small.  We took water samples at these depths just in case the signal might be real.  We also took our standard upper water column samples (described above).  Again, there are very few plankton in the samples.

1918 CDT arrive at S9,T22 for CTD cast.  No significant Aquatracka signal, but we did take a sample in an area of small increase to groundtruth the signal.  Also, the signal was no longer noisy in the upper water column indicating that sunlight may affect the instrument in the upper water column.  We also took our standard upper water column samples and again, there were few plankton in the water.

10:10 CDT arrive at station S10,T22 to conduct CTD.  There was no significant signal in the Aquatracka signal.  

After processing the samples at this station, we broke off operations to transit back to B246 for a 24 hour time series of CTD and MOCNESS.

9/8/10

0700 CDT we conducted a CTD cast.  There was a tiny signal in the Aquatracka at 620 m, but no DO sag.  We took a water sample just to groundtruth this small signal.  We also took our standard upper water column samples.  We can see Trichodesmuim colonies in the water by observing from deck, but there are very few plankton in the samples.

0936 MOCNESS tow.  We launched the MOC and did a tow that began at 800 m.  We followed the predeploymnet design of 200 m depth strata until the upper 200 m where we took a sample from 200-25 m and 25-0 m.  The angle reading of the net was always 0 on the cast and very suspicious.  We also lost the Flowmeter readings during the tow with net 3 open (600-400 m).  We completed the cast and recorded the average observed water filtered over time, etc. so that we can estimate volume filtered later.

The samples appeared to be in good condition and contained some smaller jellies, shrimp, small fish and larval fish, copepods, chaetognaths, and other small plankton.

We are remaining on station to conduct night operations in the same location.

